Author’s response to reviews

Title: Isolated loss of inferior pubic ramus: a case report

Authors:

Aly Saber (alysaber@hotmail.com)

Version: 5 Date: 26 March 2008

Author's response to reviews:

Revised points

1- A fit and otherwise healthy 35 year old female presented” rather than “a young adult female, 35 years old with normal daily activity” line 12 under: Case presentation

2- A bit more information on the embryological development of the pubis is added as paragraphs 3 and 4 under: Introduction

3- As regard to the knee problem, the following paragraph is added under: Case presentation:
She was complaining of pain in the left hip joint that got worse by walking and long standing, otherwise there was no complaint regarding both knees.

4- At the end of Conclusion, the following statement is added: The answer is that we don’t really know what has caused this abnormality.

5- Ossification centers instead of ossification centres in line 6 under: Introduction